% $Id: paper.tex,v 1.105 2018/12/21 19:19:40 brandenb Exp $
\documentclass[iop,numberedappendix]{emulateapj-rtx4}
\usepackage{graphicx,natbib,color,bm,url,times}
\topmargin 0mm %(for letter paper format)
\graphicspath{{./fig/}{./png/}}
%\def\red{\textcolor{red}}
%\def\blue{\textcolor{blue}}

%|||||||||||||||||||||||||||||||||||||||||||||||||||||||||||||||||||
%             Customized Commands
%|||||||||||||||||||||||||||||||||||||||||||||||||||||||||||||||||||
%  mathematical abbreviations
%  =========================

% math defs
\newcommand{\EQ}{\begin{equation}}
\newcommand{\EN}{\end{equation}}
\newcommand{\EQA}{\begin{eqnarray}}
\newcommand{\ENA}{\end{eqnarray}}
\newcommand{\eq}[1]{(\ref{#1})}
\newcommand{\EEq}[1]{Equation~(\ref{#1})}
%\newcommand{\Eq}[1]{Eq.~(\ref{#1})}
%\newcommand{\Eqs}[2]{Eqs.~(\ref{#1}) and~(\ref{#2})}
%\newcommand{\Eqss}[2]{Eqs.~(\ref{#1})--(\ref{#2})}
\newcommand{\Eq}[1]{Equation~(\ref{#1})}
\newcommand{\Eqs}[2]{Equations~(\ref{#1}) and~(\ref{#2})}
\newcommand{\Eqss}[2]{Equations~(\ref{#1})--(\ref{#2})}
\newcommand{\eqs}[2]{(\ref{#1}) and~(\ref{#2})}
\newcommand{\App}[1]{Appendix~\ref{#1}}
%\newcommand{\Sec}[1]{Sect.~\ref{#1}}
%\newcommand{\Secs}[2]{Sects.~\ref{#1} and \ref{#2}}
\newcommand{\Sec}[1]{Section~\ref{#1}}
\newcommand{\Secs}[2]{Sections~\ref{#1} and \ref{#2}}

%\newcommand{\Fig}[1]{Fig.~\ref{#1}}
%\newcommand{\FFig}[1]{Figure~\ref{#1}}
%\newcommand{\Figp}[2]{Fig.~\ref{#1}({#2})}
%\newcommand{\Figsp}[3]{Figs.~\ref{#1}({#2}) and ({#3})}
%\newcommand{\Figssp}[3]{Figs.~\ref{#1}({#2})--({#3})}
%\newcommand{\Figs}[2]{Figs.~\ref{#1} and \ref{#2}}
%\newcommand{\Figss}[2]{Figs.~\ref{#1}--\ref{#2}}

\newcommand{\Fig}[1]{Figure~\ref{#1}}
\newcommand{\FFig}[1]{Figure~\ref{#1}}
\newcommand{\Figp}[2]{Figure~\ref{#1}({#2})}
\newcommand{\Figsp}[3]{Figures~\ref{#1}({#2}) and ({#3})}
\newcommand{\Figssp}[3]{Figures~\ref{#1}({#2})--({#3})}
\newcommand{\Figs}[2]{Figures~\ref{#1} and \ref{#2}}
\newcommand{\Figss}[2]{Figures~\ref{#1}--\ref{#2}}
\newcommand{\Tab}[1]{Table~\ref{#1}}
\newcommand{\Tabs}[2]{Tables~\ref{#1} and \ref{#2}}

%\newcommand{\Fig}[1]{\textcolor{red}{Figure~\ref{#1}}}
%\newcommand{\FFig}[1]{\textcolor{red}{Figure~\ref{#1}}}
%\newcommand{\Figs}[2]{\textcolor{red}{Figures~\ref{#1} and \ref{#2}}}
%\newcommand{\Figss}[2]{\textcolor{red}{Figures~\ref{#1}--\ref{#2}}}
%\newcommand{\Figp}[2]{\textcolor{red}{Figure~\ref{#1}({#2})}}
%\newcommand{\Figsp}[3]{\textcolor{red}{Figures~\ref{#1}({#2}) and ({#3})}}
%\newcommand{\Figssp}[3]{\textcolor{red}{Figures~\ref{#1}({#2})--({#3})}}
%\newcommand{\Tab}[1]{\textcolor{magenta}{Table~\ref{#1}}}
%\newcommand{\Tabs}[2]{\textcolor{red}{Tables~\ref{#1} and \ref{#2}}}

\newcommand{\bra}[1]{\langle #1\rangle}
\newcommand{\bbra}[1]{\left\langle #1\right\rangle}
\newcommand{\mean}[1]{\overline #1}
\newcommand{\meanrho}{\overline{\rho}}
\newcommand{\meanPhi}{\overline{\Phi}}
\newcommand{\tildeFFFF}{\tilde{\mbox{\boldmath ${\cal F}$}}{}}{}
\newcommand{\hatFFFF}{\hat{\mbox{\boldmath ${\cal F}$}}{}}{}
\newcommand{\meanFFFF}{\overline{\mbox{\boldmath ${\cal F}$}}{}}{}
\newcommand{\meanemf}{\overline{\cal E} {}}
\newcommand{\meanAAAA}{\overline{\mbox{\boldmath ${\mathsf A}$}} {}}
\newcommand{\meanSSSS}{\overline{\mbox{\boldmath ${\mathsf S}$}} {}}
\newcommand{\meanAAA}{\overline{\mathsf{A}}}
\newcommand{\meanSSS}{\overline{\mathsf{S}}}
\newcommand{\meanuu}{\overline{\mbox{\boldmath $u$}}{}}{}
\newcommand{\meanoo}{\overline{\mbox{\boldmath $\omega$}}{}}{}
\newcommand{\meanEMF}{\overline{\mbox{\boldmath ${\cal E}$}}{}}{}
\newcommand{\meanEEEE}{\overline{\mbox{\boldmath ${\cal E}$}}{}}{}
\newcommand{\meanuxB}{\overline{\mbox{\boldmath $\delta u\times \delta B$}}{}}{}
\newcommand{\meanJB}{\overline{\mbox{\boldmath $J\cdot B$}}{}}{}
\newcommand{\meanAB}{\overline{\mbox{\boldmath $A\cdot B$}}{}}{}
\newcommand{\meanjb}{\overline{\mbox{\boldmath $j\cdot b$}}{}}{}
\newcommand{\meanEE}{\overline{\mbox{\boldmath $E$}}{}}{}
\newcommand{\meanFFf}{\overline{\mbox{\boldmath $F$}}_{\rm f}{}}{}
\newcommand{\meanFFm}{\overline{\mbox{\boldmath $F$}}_{\rm m}{}}{}
\newcommand{\hatFF}{\hat{\mbox{\boldmath $F$}}{}}{}
\newcommand{\meanGG}{\overline{\mbox{\boldmath $G$}}{}}{}
\newcommand{\meanHH}{\overline{\mbox{\boldmath $H$}}{}}{}
\newcommand{\tildeGG}{\tilde{\mbox{\boldmath $G$}}{}}{}
\newcommand{\meanCC}{\overline{\mbox{\boldmath $C$}}{}}{}
\newcommand{\meanKK}{\overline{\mbox{\boldmath $K$}}{}}{}
\newcommand{\meanUU}{\overline{\bm{U}}}
\newcommand{\meanVV}{\overline{\bm{V}}}
\newcommand{\meanWW}{\overline{\bm{W}}}
\newcommand{\meanQQ}{\overline{\mbox{\boldmath $Q$}}{}}{}
\newcommand{\meana}{\overline{a}}
\newcommand{\meanA}{\overline{A}}
\newcommand{\meanB}{\overline{B}}
\newcommand{\meanE}{\overline{E}}
\newcommand{\meanK}{\overline{K}}
\newcommand{\meanb}{\tilde{b}}
\newcommand{\meanj}{\tilde{j}}
\newcommand{\meanC}{\overline{C}}
\newcommand{\meanD}{\overline{D}}
\newcommand{\meanF}{\overline{F}}
\newcommand{\meanFm}{\overline{F}_{\rm m}}
\newcommand{\meanFf}{\overline{F}_{\rm f}}
\newcommand{\meanh}{\overline{h}}
\newcommand{\meanhm}{\overline{h}_{\rm m}}
\newcommand{\meanhf}{\overline{h}_{\rm f}}
\newcommand{\meanG}{\overline{G}}
\newcommand{\meanGGG}{\overline{\cal G}}
\newcommand{\meanH}{\overline{H}}
\newcommand{\meanU}{\overline{U}}
\newcommand{\meanR}{\overline{R}}
\newcommand{\meanJ}{\overline{J}}
\newcommand{\means}{\overline{s}}
\newcommand{\meanp}{\overline{p}}
\newcommand{\meanP}{\overline{P}}
\newcommand{\meanS}{\overline{S}}
\newcommand{\meanT}{\overline{T}}
\newcommand{\meanW}{\overline{W}}
\newcommand{\meanQ}{\overline{Q}}
\newcommand{\meanEEE}{\overline{\cal E}}
\newcommand{\meanFFF}{\overline{\cal F}}
\newcommand{\meanHHH}{\overline{\cal H}}
\newcommand{\meanRRR}{\overline{\cal R}}
%\newcommand{\hatAA}{\hat{\mbox{\boldmath $A$}}{}}{}
%\newcommand{\hatBB}{\hat{\mbox{\boldmath $B$}}{}}{}
\newcommand{\hatg}{\hat{g}}
\newcommand{\hatk}{\hat{k}}
\newcommand{\hatkk}{\hat{\bm{k}}}
\newcommand{\hatAA}{\hat{\bm{A}}}
\newcommand{\hatBB}{\hat{\bm{B}}}
\newcommand{\hatJJ}{\hat{\mbox{\boldmath $J$}}{}}{}
%\newcommand{\hatOO}{\hat{\mbox{\boldmath $\Omega$}} {}}
\newcommand{\hatOO}{\hat{\bm{\Omega}}}
\newcommand{\hatEMF}{\hat{\mbox{\boldmath ${\cal E}$}}{}}{}
\newcommand{\Rgas}{{\cal R}}{}
\newcommand{\emf}{{\cal E}}{}
\newcommand{\hatA}{\hat{A}}
\newcommand{\hatB}{\hat{B}}
\newcommand{\hatJ}{\hat{J}}
\newcommand{\hatU}{\hat{U}}
\newcommand{\alphaK}{\alpha_{\rm K}}
\newcommand{\alphaM}{\alpha_{\rm M}}
\newcommand{\alphaSS}{\alpha_{\rm SS}}
%
% tilde
%
\newcommand{\tildeB}{\tilde{B}}
\newcommand{\tildeJ}{\tilde{J}}
\newcommand{\tildeemf}{\tilde{\cal E}}
\newcommand{\teps}{\tilde{\epsilon} {}}
\newcommand{\Oh}{\hat{\Omega}}
\newcommand{\zh}{\hat{z}}
\newcommand{\PC}{{\sc Pencil Code}~}
%
%  unit vectors
%
\newcommand{\pphi}{\hat{\bm{\phi}}}
\newcommand{\ppom}{\bm{\hat{\varpi}}}
%\newcommand{\ppom}{\hat{\bm{\varpi}}}
\newcommand{\eee}{\hat{\mbox{\boldmath $e$}} {}}
%\newcommand{\nnn}{\hat{\mbox{\boldmath $n$}} {}}
\newcommand{\nnn}{\hat{\bm{n}}}
\newcommand{\rrr}{\hat{\mbox{\boldmath $r$}} {}}
\newcommand{\vvv}{\hat{\mbox{\boldmath $v$}} {}}
\newcommand{\xxx}{\hat{\mbox{\boldmath $x$}} {}}
\newcommand{\yyy}{\hat{\mbox{\boldmath $y$}} {}}
\newcommand{\zzz}{\hat{\mbox{\boldmath $z$}} {}}
\newcommand{\ttt}{\hat{\mbox{\boldmath $\theta$}} {}}
\newcommand{\OOO}{\hat{\mbox{\boldmath $\Omega$}} {}}
\newcommand{\ooo}{\hat{\mbox{\boldmath $\omega$}} {}}
%\newcommand{\ppom}{\hat{\mbox{\boldmath $\varpi$}} {}}
\newcommand{\BBBB}{\hat{\mbox{\boldmath $B$}} {}}
\newcommand{\BBhat}{\hat{\bm{B}}}
\newcommand{\meanAA}{{\overline{\bm{A}}}}
\newcommand{\meanBB}{{\overline{\bm{B}}}}
\newcommand{\meanJJ}{{\overline{\bm{J}}}}
\newcommand{\meanFF}{{\overline{\bm{F}}}}
\newcommand{\meanBBhat}{\hat{\overline{\bm{B}}}}
\newcommand{\meanJJhat}{\hat{\overline{\bm{J}}}}
\newcommand{\meanBhat}{\hat{\overline{B}}}
\newcommand{\meanJhat}{\hat{\overline{J}}}
\newcommand{\Bhat}{\hat{B}}
\newcommand{\Bh}{\hat{B}}
%
%  vectors
%
\newcommand{\nullvector}{{\bf0}}
\newcommand{\gggg}{\mbox{\boldmath $g$} {}}
\newcommand{\ddd}{\mbox{\boldmath $d$} {}}
\newcommand{\yy}{\mbox{\boldmath $y$} {}}
\newcommand{\zz}{\mbox{\boldmath $z$} {}}
\newcommand{\vv}{\mbox{\boldmath $v$} {}}
\newcommand{\ww}{\mbox{\boldmath $w$} {}}
\newcommand{\mm}{\mbox{\boldmath $m$} {}}
\newcommand{\PP}{\mbox{\boldmath $P$} {}}
\newcommand{\bp}{\mbox{\boldmath $p$} {}}
\newcommand{\II}{\mbox{\boldmath $I$} {}}
\newcommand{\RR}{\mbox{\boldmath $R$} {}}
\newcommand{\kk}{\bm{k}}
\newcommand{\KK}{\bm{K}}
\newcommand{\pp}{\bm{p}}
\newcommand{\qq}{\bm{q}}
\newcommand{\xx}{\bm{x}}
\newcommand{\XX}{\bm{X}}
\newcommand{\aaaa}{\bm{a}}
\newcommand{\bb}{\bm{b}}
\newcommand{\jj}{\bm{j}}
\newcommand{\rr}{\bm{r}}
\newcommand{\grav}{\bm{g}}
\newcommand{\BB}{\bm{B}}
\newcommand{\EE}{\bm{E}}
\newcommand{\JJ}{\bm{J}}
\newcommand{\oo}{\bm{\omega}}
\newcommand{\AAA}{\bm{A}}
\newcommand{\HH}{\bm{H}}
\newcommand{\MM}{\bm{M}}
\newcommand{\UU}{\bm{U}}
\newcommand{\FF}{\bm{F}}
\def\os{\omega_{\rm c}}
\def\oA{\omega_{\rm A}}
\def\oAy{\omega_{{\rm A}y}}
\def\oAz{\omega_{{\rm A}z}}
\newcommand{\vvA}{\bm{v}_{\rm A}}
\newcommand{\uu}{\bm{u}}
\newcommand{\uud}{\bm{u}_{\rm d}}
\newcommand{\rhod}{\rho_{\rm d}}
\newcommand{\sss}{\mbox{\boldmath $s$} {}}
\newcommand{\SSS}{\mbox{\boldmath $S$} {}}
\newcommand{\ee}{\mbox{\boldmath $e$} {}}
\newcommand{\nn}{\mbox{\boldmath $n$} {}}
\newcommand{\ff}{\mbox{\boldmath $f$} {}}
\newcommand{\hh}{\mbox{\boldmath $h$} {}}
\newcommand{\EEE}{\mbox{\boldmath ${\cal E}$} {}}
\newcommand{\FFF}{\mbox{\boldmath ${\cal F}$} {}}
\newcommand{\TT}{{\bm{T}}}
\newcommand{\nab}{{\bm{\nabla}}}
\newcommand{\GG}{\mbox{\boldmath $G$} {}}
\newcommand{\WW}{\mbox{\boldmath $W$} {}}
\newcommand{\QQ}{\mbox{\boldmath $Q$} {}}
%\newcommand{\ggg}{\mbox{\boldmath $g$} {}}
%\newcommand{\nab}{\mbox{\boldmath $\nabla$} {}}
\newcommand{\nabD}{\nabla_{\rm D}}
\newcommand{\nabel}{\nabla_{\rm el}}
\newcommand{\nabad}{\nabla_{\rm ad}}
\newcommand{\nabrad}{\nabla_{\rm rad}}
\newcommand{\nabk}{\mbox{\boldmath $\nabla$}_{\!\kk\,}}
%\newcommand{\OO}{\mbox{\boldmath $\Omega$} {}}
\newcommand{\OO}{\bm{\Omega}}
\newcommand{\ppsi}{\mbox{\boldmath $\psi$} {}}
%\newcommand{\oo}{\mbox{\boldmath $\omega$} {}}
\newcommand{\pom}{\mbox{\boldmath $\varpi$} {}}
\newcommand{\ttau}{\mbox{\boldmath $\tau$} {}}
\newcommand{\LL}{\mbox{\boldmath $\Lambda$} {}}
\newcommand{\mmu}{\mbox{\boldmath $\mu$} {}}
\newcommand{\xxi}{\mbox{\boldmath $\xi$} {}}
\newcommand{\ddelta}{\mbox{\boldmath $\delta$} {}}
\newcommand{\ggamma}{\mbox{\boldmath $\gamma$} {}}
\newcommand{\kkappa}{\mbox{\boldmath $\kappa$} {}}
\newcommand{\llambda}{\mbox{\boldmath $\lambda$} {}}
\newcommand{\pomega}{\mbox{\boldmath $\varpi$} {}}
\newcommand{\PPsi}{\mbox{\boldmath $\Psi$} {}}
%
%  tensors
%
\newcommand{\DDDD}{\mbox{\boldmath ${\sf D}$} {}}
\newcommand{\IIII}{\mbox{\boldmath ${\sf I}$} {}}
\newcommand{\LLLL}{\mbox{\boldmath ${\sf L}$} {}}
\newcommand{\MMMM}{\mbox{\boldmath ${\sf M}$} {}}
\newcommand{\NNNN}{\mbox{\boldmath ${\sf N}$} {}}
\newcommand{\PPPP}{\mbox{\boldmath ${\sf P}$} {}}
\newcommand{\QQQQ}{\mbox{\boldmath ${\sf Q}$} {}}
\newcommand{\RRRR}{\mbox{\boldmath ${\sf R}$} {}}
\newcommand{\SSSS}{\mbox{\boldmath ${\sf S}$} {}}
\newcommand{\BBBBB}{\mbox{\boldmath ${\sf B}$} {}}
\newcommand{\tAAAA}{\tilde{\mbox{\boldmath ${\sf A}$}} {}}
\newcommand{\tDDDD}{\tilde{\mbox{\boldmath ${\sf D}$}} {}}
\newcommand{\tRRRR}{\tilde{\mbox{\boldmath ${\sf R}$}} {}}
\newcommand{\tQQQQ}{\tilde{\mbox{\boldmath ${\sf Q}$}} {}}
%
\newcommand{\AAAA}{\mbox{\boldmath ${\cal A}$} {}}
\newcommand{\BBB}{\mbox{\boldmath ${\cal B}$} {}}
\newcommand{\EMF}{\mbox{\boldmath ${\cal E}$} {}}
\newcommand{\GGG}{\mbox{\boldmath ${\cal G}$} {}}
\newcommand{\HHH}{\mbox{\boldmath ${\cal H}$} {}}
%\newcommand{\QQQ}{\mbox{\boldmath ${\cal Q}$} {}}
\newcommand{\meanQQQ}{\overline{\mbox{\boldmath ${\cal Q}$}} {}}
%
\newcommand{\GGGG}{{\bf G} {}}
%
%  operators, subscripts, etc  (roman)
%
\newcommand{\ii}{{\rm i}}
\newcommand{\erf}{{\rm erf}}
\newcommand{\VAR}{{\rm var}}
\newcommand{\SKEW}{{\rm skew}}
\newcommand{\KURT}{{\rm kurt}}
\newcommand{\grad}{{\rm grad} \, {}}
\newcommand{\curl}{{\rm curl} \, {}}
\newcommand{\dive}{{\rm div}  \, {}}
\newcommand{\Dive}{{\rm Div}  \, {}}
\newcommand{\diag}{{\rm diag}  \, {}}
\newcommand{\asin}{{\rm asin}  \, {}}
\newcommand{\sgn}{{\rm sgn}  \, {}}
\newcommand{\meanDD}{{\overline{\rm D}} {}}
\newcommand{\DD}{{\rm D} {}}
\newcommand{\DDD}{{\cal D} {}}
\newcommand{\QQQ}{{\cal Q}}
\newcommand{\dd}{{\rm d} {}}
\newcommand{\dV}{\,{\rm d}V {}}
\newcommand{\dS}{\,{\rm d}{{\bm{S}}} {}}
\newcommand{\const}{{\rm const}  {}}
\newcommand{\CR}{{\rm CR}}
\def\degr{\hbox{$^\circ$}}
\def\la{\mathrel{\mathchoice {\vcenter{\offinterlineskip\halign{\hfil
$\displaystyle##$\hfil\cr<\cr\sim\cr}}}
{\vcenter{\offinterlineskip\halign{\hfil$\textstyle##$\hfil\cr<\cr\sim\cr}}}
{\vcenter{\offinterlineskip\halign{\hfil$\scriptstyle##$\hfil\cr<\cr\sim\cr}}}
{\vcenter{\offinterlineskip\halign{\hfil$\scriptscriptstyle##$\hfil\cr<\cr\sim\cr}}}}}
\def\ga{\mathrel{\mathchoice {\vcenter{\offinterlineskip\halign{\hfil
$\displaystyle##$\hfil\cr>\cr\sim\cr}}}
{\vcenter{\offinterlineskip\halign{\hfil$\textstyle##$\hfil\cr>\cr\sim\cr}}}
{\vcenter{\offinterlineskip\halign{\hfil$\scriptstyle##$\hfil\cr>\cr\sim\cr}}}
{\vcenter{\offinterlineskip\halign{\hfil$\scriptscriptstyle##$\hfil\cr>\cr\sim\cr}}}}}
%
%  numbers
%
\def\Ta{\mbox{\rm Ta}}
\def\Ra{\mbox{\rm Ra}}
\def\Rat{\mbox{\rm Ra}_{\rm turb}}
\def\Prt{\mbox{\rm Pr}_{\rm turb}}
\def\Ma{\mbox{\rm Ma}}
\def\Co{\mbox{\rm Co}}
\def\Ro{\mbox{\rm Ro}}
\def\Ri{\mbox{\rm Ri}}
\def\CoM{\mbox{\rm Co}_{\rm M}}
\def\Sh{\mbox{\rm Sh}}
\def\St{\mbox{\rm St}}
\def\Roo{\mbox{\rm Ro}^{-1}}
\def\Rooo{\mbox{\rm Ro}^{-2}}
\def\Pra{\mbox{\rm Pr}}
\def\Pran{\mbox{\rm Pr}}
\def\Sc{\mbox{\rm Sc}}
\def\Tr{\mbox{\rm Tr}}
\def\Da{\mbox{\rm Da}}
\def\Ka{\mbox{\rm Ka}}
\def\RM{\mbox{\rm RM}}
\def\Pm{\mbox{\rm Pr}_M}
%\def\Rm{\mbox{\rm Re}_M}
\def\Rm{R_{\rm m}}
\def\Dt{D_{\rm t}}
\def\DT{D_{\rm T}}
%\def\Rmc{\mbox{\rm Re}_{M,{\rm crit}}}
%\def\Pm{P_{\rm m}}
%\def\Rm{R_{\rm m}}
\def\Rmc{R_{\rm m,{\rm crit}}}
\def\Rey{\mbox{\rm Re}}
\def\Imag{\mbox{\rm Im}}
\def\Pe{\mbox{\rm Pe}}
\def\Co{\mbox{\rm Co}}
\def\Gr{\mbox{\rm Gr}}
\def\Lu{\mbox{\rm Lu}}
\def\tauc{\tau_{\rm c}}
\def\tauA{\tau_{\rm A}}
\def\taueff{\tau_{\rm eff}}
\def\HP{H_{\!P}}
\def\cP{c_{P}}
\def\cV{c_{V}}
\def\alpK{\alpha_{\rm K}}
\def\alpM{\alpha_{\rm M}}
\def\EEK{{\cal E}_{\rm K}}
\def\EEM{{\cal E}_{\rm M}}
\def\EK{E_{\rm K}}
\def\EM{E_{\rm M}}
\def\EKM{E_{\rm K/M}}
\def\ET{E_{\rm T}}
\def\WL{W_{\rm L}}
\def\WC{W_{\rm C}}
\def\cmag{c_{\rm m}}
\def\cmso{c_{\rm ms}}
\def\csz{c_{\rm s0}}
\def\Teff{T_{\rm eff}}
\def\cp{c_{\rm p}}
\def\cv{c_{\rm v}}
\def\Rs{R_{\odot}}
\def\cs{c_{\rm s}}
\def\xiM{\xi_{\rm M}}
\def\xiK{\xi_{\rm K}}
\def\xiKM{\xi_{\rm K/M}}
\def\Hpz{H_{\rm p0}}
\def\Hrz{H_{\rho 0}}
\def\qpz{q_{\rm p0}}
\def\qp{q_{\rm p}}
\def\betaK{\beta_{\rm K}}
\def\betaM{\beta_{\rm M}}
\def\betaKM{\beta_{\rm K/M}}
\def\betap{\beta_{\rm p}}
\def\betamin{\beta_{\min}}
\def\betastar{\beta_{\star}}
\def\Peff{{\cal P}_{\rm eff}}
\def\Pmin{{\cal P}_{\rm min}}
\def\pturbz{p_{\rm turb}^{(0)}}
\def\pturb{p_{\rm turb}}
\def\peff{p_{\rm eff}}
\def\qs{q_{\rm s}}
\def\qe{q_{\rm e}}
\def\qg{q_{\rm g}}
\def\qa{q_{\rm a}}
\def\pt{p_{\rm t}}
\def\ptz{p_{\rm t0}}
\def\vA{v_{\rm A}}
\def\vAy{v_{{\rm A}y}}
\def\vAZ{v_{{\rm A}z}}
\def\vAz{v_{\rm A0}}
\def\hf{h_{\rm f}}
\def\hm{h_{\rm m}}
\def\hrms{h_{\rm rms}}
\def\kmean{k_{\rm m}}
\def\lf{l_{\rm f}}
\def\kone{k_1}
\def\kd{k_{\rm d}}
\def\kf{k_{\rm f}}
\def\kp{k_{\rm p}}
\def\Hp{H_{\rm p}}
\def\HM{H_{\rm M}}
\def\EM{E_{\rm M}}
\def\Ekin{E_{\rm kin}}
\def\kB{k_{\rm B}}
\def\sigmaSB{\sigma_{\rm SB}}
\def\kappaf{\kappa_{\rm f}}
\def\epsf{\epsilon_{\rm f}}
\def\epsfz{\epsilon_{\rm f0}}
\def\epsK{\epsilon_{\it K}}
\def\epsM{\epsilon_{\it M}}
\def\epsC{\epsilon_{\it C}}
\def\epsT{\epsilon_{\it T}}
\def\betK{\beta_{\it K}}
\def\betM{\beta_{\it M}}
\def\xid{\xi_{\rm D}}
\def\NA{N_{\rm A}}
\def\kB{k_{\rm B}}
\def\kK{k_{\it K}}
\def\kM{k_{\it M}}
\def\kmax{k_{\rm max}}
\def\kom{k_{\omega}}
\def\Ff{F_{\rm f}}
\def\Fm{F_{\rm m}}
\def\Hf{H_{\rm f}}
\def\Hm{H_{\rm m}}
\def\vArms{v_{\rm A,rms}}
\def\Brms{B_{\rm rms}}
\def\Jrms{J_{\rm rms}}
\def\Urms{U_{\rm rms}}
\def\urms{u_{\rm rms}}
\def\wrms{w_{\rm rms}}
\def\orms{\omega_{\rm rms}}
\def\omms{\omega_{\rm ms}}
\def\brms{b_{\rm rms}}
\def\uref{u_{\rm ref}}
\def\yN{y_{\rm N}}
\def\yL{y_{\rm L}}
\def\kappaBB{\kappa_{\rm BB}}
\def\kappaB{\kappa_{\rm B}}
\def\kappaOO{\kappa_{\Omega\Omega}}
\def\kappaO{\kappa_{\Omega}}
\def\kappah{\kappa_{\rm h}}
\def\kappat{\kappa_{\rm t}}
\def\kappaT{\kappa_{\rm T}}
\def\kappatz{\kappa_{\rm t0}}
\def\mut{\mu_{\rm t}}
\def\nut{\nu_{\rm t}}
\def\nuM{\nu_{\rm M}}
\def\nuT{\nu_{\rm T}}
\def\calKt{{\cal K}_{\rm t}}
\def\calKtz{{\cal K}_{\rm t0}}
\def\chit{\chi_{\rm t}}
\def\chitz{\chi_{\rm t0}}
\def\etat{\eta_{\rm t}}
\def\etatz{\eta_{\rm t0}}
\def\etatz{\eta_{\rm t0}}
\def\etaTz{\eta_{\rm T0}}
\def\nutz{\nu_{\rm t0}}
\def\etaT{\eta_{\rm T}}
\def\mut{\mu_{\rm t}}
\def\muT{\mu_{\rm T}}
\def\uT{\mu{\rm T}}
\def\BBeq{|\BB|/B_{\rm eq}}
\def\Beq{B_{\rm eq}}
\def\Beqz{B_{\rm eq0}}
\def\tautd{\tau_{\rm td}}
\def\tauto{\tau_{\rm to}}
%
\newcommand{\ea}{{\rm et al.\ }}
\newcommand{\eaa}{{\rm et al.\ }}
\def\half{{\textstyle{1\over2}}}
\def\hhalf{{\scriptstyle{1\over2}}}
\def\threehalf{{\textstyle{3\over2}}}
\def\threequarter{{\textstyle{3\over4}}}
\def\sevenhalf{{\textstyle{7\over2}}}
\def\onethird{{\textstyle{1\over3}}}
\def\onesixth{{\textstyle{1\over6}}}
\def\twothird{{\textstyle{2\over3}}}
\def\fivethird{{\textstyle{5\over3}}}
\def\onenineth{{\textstyle{1\over9}}}
\def\fourthird{{\textstyle{4\over3}}}
\def\quarter{{\textstyle{1\over4}}}
%
\newcommand{\W}{\,{\rm W}}
\newcommand{\V}{\,{\rm V}}
\newcommand{\kV}{\,{\rm kV}}
\newcommand{\T}{\,{\rm T}}
\newcommand{\G}{\,{\rm G}}
\newcommand{\sfu}{\,{\rm sfu}}
\newcommand{\Jy}{\,{\rm Jy}}
\newcommand{\Hz}{\,{\rm Hz}}
\newcommand{\nHz}{\,{\rm nHz}}
\newcommand{\GHz}{\,{\rm GHz}}
\newcommand{\kHz}{\,{\rm kHz}}
\newcommand{\kG}{\,{\rm kG}}
\newcommand{\K}{\,{\rm K}}
\newcommand{\g}{\,{\rm g}}
\newcommand{\s}{\,{\rm s}}
\newcommand{\um}{\,\mu{\rm m}}
\newcommand{\ms}{\,{\rm ms}}
\newcommand{\mpers}{\,{\rm m/s}}
\newcommand{\ks}{\,{\rm ks}}
\newcommand{\cm}{\,{\rm cm}}
\newcommand{\mV}{\,{\rm mV}}
\newcommand{\cmcube}{\,{\rm cm^{-3}}}
\newcommand{\nm}{\,{\rm nm}}
\newcommand{\m}{\,{\rm m}}
\newcommand{\km}{\,{\rm km}}
\newcommand{\msec}{\,{\rm ms}}
\newcommand{\cms}{\,{\rm cm/s}}
\newcommand{\kms}{\,{\rm km/s}}
\newcommand{\gpercc}{\,{\rm g/cm}^3}
\newcommand{\kg}{\,{\rm kg}}
\newcommand{\ug}{\,\mu{\rm g}}
\newcommand{\kW}{\,{\rm kW}}
\newcommand{\MW}{\,{\rm MW}}
\newcommand{\Mm}{\,{\rm Mm}}
\newcommand{\Mx}{\,{\rm Mx}}
\newcommand{\pc}{\,{\rm pc}}
\newcommand{\kpc}{\,{\rm kpc}}
\newcommand{\yr}{\,{\rm yr}}
\newcommand{\Myr}{\,{\rm Myr}}
\newcommand{\Gyr}{\,{\rm Gyr}}
\newcommand{\erg}{\,{\rm erg}}
\newcommand{\mol}{\,{\rm mol}}
\newcommand{\dyn}{\,{\rm dyn}}
\newcommand{\J}{\,{\rm J}}
%\newcommand{\RM}{\,{\rm RM}}
%\newcommand{\EM}{\,{\rm EM}}
\newcommand{\AU}{\,{\rm AU}}
\newcommand{\A}{\,{\rm A}}
%\newcommand{\kA}{\,{\rm kA}}
\newcommand{\etal}{et al.}
\newcommand{\Hminus}{{\rm H}^{-}}
\newcommand{\Mdot}{\dot{M}}
\newcommand{\MSun}{M_\odot}
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%  journals
%
\newcommand{\yastroph}[2]{ #1, astro-ph/#2}
\newcommand{\ycsf}[3]{ #1, {Chaos, Solitons \& Fractals,} {#2}, #3}
\newcommand{\yepl}[3]{ #1, {Europhys.\ Lett.,} {#2}, #3}
\newcommand{\yaj}[3]{ #1, {AJ,} {#2}, #3}
\newcommand{\yjgr}[3]{ #1, {J.\ Geophys.\ Res.,} {#2}, #3}
\newcommand{\yjas}[3]{ #1, {J.\ Atmos.\ Sci.,} {#2}, #3}
\newcommand{\ysol}[3]{ #1, {Sol.\ Phys.,} {#2}, #3}
\newcommand{\ymonber}[3]{ #1, {Monats.\ Dt.\ Akad.\ Wiss.,} {#2}, #3}
\newcommand{\yapj}[3]{ #1, {ApJ,} {#2}, #3}
\newcommand{\ypasp}[3]{ #1, {PASP,} {#2}, #3}
\newcommand{\yapjl}[3]{ #1, {ApJ,} {#2}, #3}
\newcommand{\yapjs}[3]{ #1, {ApJS,} {#2}, #3}
\newcommand{\yija}[3]{ #1, {Int.\ J.\ Astrobiol.,} {#2}, #3}
\newcommand{\yan}[3]{ #1, {Astron.\ Nachr.,} {#2}, #3}
\newcommand{\yzfa}[3]{ #1, {Z.\ f.\ Ap.,} {#2}, #3}
\newcommand{\ymhdn}[3]{ #1, {Magnetohydrodyn.} {#2}, #3}
\newcommand{\yana}[3]{ #1, {A\&A,} {#2}, #3}
\newcommand{\yanas}[3]{ #1, {A\&AS,} {#2}, #3}
\newcommand{\yanar}[3]{ #1, {A\&A Rev.,} {#2}, #3}
\newcommand{\yass}[3]{ #1, {Ap\&SS,} {#2}, #3}
\newcommand{\ygafd}[3]{ #1, {Geophys.\ Astrophys.\ Fluid Dyn.,} {#2}, #3}
\newcommand{\ygrl}[3]{ #1, {Geophys.\ Res.\ Lett.,} {#2}, #3}
\newcommand{\ypasj}[3]{ #1, {Publ.\ Astron.\ Soc.\ Japan,} {#2}, #3}
\newcommand{\yjfm}[3]{ #1, {J.\ Fluid Mech.,} {#2}, #3}
\newcommand{\ypepi}[3]{ #1, {Phys.\ Earth Planet.\ Int.,} {#2}, #3}
\newcommand{\ypf}[3]{ #1, {Phys.\ Fluids,} {#2}, #3}
\newcommand{\ypfb}[3]{ #1, {Phys.\ Fluids B,} {#2}, #3}
\newcommand{\ypp}[3]{ #1, {Phys.\ Plasmas,} {#2}, #3}
\newcommand{\ysov}[3]{ #1, {Sov.\ Astron.,} {#2}, #3}
\newcommand{\ysovl}[3]{ #1, {Sov.\ Astron.\ Lett.,} {#2}, #3}
\newcommand{\yjetp}[3]{ #1, {Sov.\ Phys.\ JETP,} {#2}, #3}
\newcommand{\yphy}[3]{ #1, {Physica,} {#2}, #3}
\newcommand{\yaraa}[3]{ #1, {ARA\&A,} {#2}, #3}
\newcommand{\yanf}[3]{ #1, {Ann. Rev. Fluid Mech.,} {#2}, #3}
\newcommand{\yrpp}[3]{ #1, {Rep.\ Prog.\ Phys.,} {#2}, #3}
\newcommand{\yprs}[3]{ #1, {Proc.\ Roy.\ Soc.\ Lond.,} {#2}, #3}
\newcommand{\yprt}[3]{ #1, {Phys.\ Rep.,} {#2}, #3}
\newcommand{\yprl}[3]{ #1, {Phys.\ Rev.\ Lett.,} {#2}, #3}
\newcommand{\yphl}[3]{ #1, {Phys.\ Lett.,} {#2}, #3}
\newcommand{\yptrs}[3]{ #1, {Phil.\ Trans.\ R.\ Soc.,} {#2}, #3}
\newcommand{\yptrsa}[3]{ #1, {Phil. Trans. R. Soc. A,} {#2}, #3}
\newcommand{\ymn}[3]{ #1, {MNRAS,} {#2}, #3}
\newcommand{\ynat}[3]{ #1, {Nature,} {#2}, #3}
\newcommand{\ysci}[3]{ #1, {Science,} {#2}, #3}
\newcommand{\ysph}[3]{ #1, {Solar Phys.,} {#2}, #3}
\newcommand{\psph}[2]{ #1, {Solar Phys.,} {#2}}
\newcommand{\yrmp}[3]{ #1, {Rev.\ Mod.\ Phys.} {#2}, #3}
\newcommand{\ypr}[3]{ #1, {Phys.\ Rev.,} {#2}, #3}
\newcommand{\ypra}[3]{ #1, {Phys.\ Rev.\ A,} {#2}, #3}
\newcommand{\yprd}[3]{ #1, {Phys.\ Rev.\ D,} {#2}, #3}
\newcommand{\ypre}[3]{ #1, {Phys.\ Rev.\ E,} {#2}, #3}
\newcommand{\ypnas}[3]{ #1, {Proc.\ Nat.\ Acad.\ Sci.,} {#2}, #3}
\newcommand{\yssr}[3]{ #1, {Spa.\ Sci.\ Rev.,} {#2}, #3}
\newcommand{\yicarus}[3]{ #1, {Icarus,} {#2}, #3}
\newcommand{\yspd}[3]{ #1, {Sov.\ Phys.\ Dokl.,} {#2}, #3}
\newcommand{\yjcp}[3]{ #1, {J.\ Comput.\ Phys.,} {#2}, #3}
\newcommand{\yjour}[4]{ #1, {#2}, {#3}, #4}
\newcommand{\yjourS}[3]{ #1, {#2}, #3}
\newcommand{\yprep}[2]{ #1, {\sf #2}}
\newcommand{\ybook}[3]{ #1, {#2} (#3)}
\newcommand{\yproc}[5]{ #1, in {#3}, ed.\ #4 (#5), #2}
\newcommand{\pproc}[4]{ #1, in {#2}, ed.\ #3 (#4), (in press)}
\newcommand{\pprocc}[5]{ #1, in {#2}, ed.\ #3 (#4, #5)}
\newcommand{\pmn}[1]{ #1, {MNRAS}, to be published}
\newcommand{\pssr}[2]{ #1, {Spa.\ Sci.\ Rev.}, in press, arXiv:#2}
\newcommand{\sana}[2]{ #1, {A\&A}, submitted, arXiv:#2}
\newcommand{\pana}[2]{ #1, {A\&A}, in press, arXiv:#2}
\newcommand{\dana}[3]{ #1, {A\&A}, DOI:#2, arXiv:#3}
\newcommand{\papj}[2]{ #1, {ApJ}, in press, arXiv:#2}
\newcommand{\pgafd}[1]{ #1, {GAFD}, to be published}
\newcommand{\ppapj}[3]{ #1, {ApJ}, {#2}, to be published in the #3 issue}
\newcommand{\sprl}[1]{ #1, {PRL}, submitted}
\newcommand{\sapj}[2]{ #1, {ApJ}, submitted, arXiv:#2}
\newcommand{\tapj}[1]{ #1, {ApJ}, to be submitted}
\newcommand{\sapjl}[2]{ #1, {ApJL}, submitted, arXiv:#2}
\newcommand{\papjl}[2]{ #1, {ApJL}, in press, arXiv:#2}
\newcommand{\ssapj}[2]{ #1, {ApJ} (#2)}
\newcommand{\san}[1]{ #1, {Astron.\ Nachr.}, submitted}
\newcommand{\ssph}[1]{ #1, {Sol.\ Phys.}, submitted}
\newcommand{\sssph}[2]{ #1, {Sol.\ Phys.} (#2)}
\newcommand{\smn}[1]{ #1, {MNRAS}, submitted}
\newcommand{\arxiv}[2]{ #1, arXiv:#2}
\hyphenation{ma-ni-fold}
\hyphenation{he-li-ci-ty}
\def\apjs{ApJS}
\def\pre{PhRvE}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\begin{document}
\begin{thebibliography}{}

\bibitem[Bai et al.(2014)]{BDTSMW14}
Bai, X. Y., Deng, Y. Y., Teng, F., Su, J. T., Mao, X. J., \& Wang, G. P.\ymn{2014}{445}{49}

\bibitem[Bourdin \& Brandenburg(2018)]{BB18}
Bourdin, Ph.-A., \& Brandenburg, A.\papj{2018}
{1804.04160}

\bibitem[Bracco et al.(2018)]{Bracco18}
Bracco, A., Candelaresi, S., Del Sordo, F., \& Brandenburg, A.\pana{2018}
{1807.10188}

\bibitem[Brandenburg(2001)]{Bra01}
Brandenburg, A.\yapj{2001}{550}{824}

\bibitem[Brandenburg \& Kahniashvili(2017)]{BK17}
Brandenburg, A., \& Kahniashvili, T.\yprl{2017}{118}{055102}

\bibitem[Brandenburg \& Subramanian(2005)]{BS05}
Brandenburg, A., \& Subramanian, K.\yjour{2005}{Phys.\ Rep.}{417}{1}

\bibitem[Brandenburg et al.(2017)]{BKMRPTV17}
Brandenburg, A., Kahniashvili, T., Mandal, S., Roper Pol, A., Tevzadze, A. G., \& Vachaspati, T.\yprd{2017}{96}{123528}

\bibitem[Brandenburg et al.(1990)]{BNPST90}
Brandenburg, A., Nordlund, \AA., Pulkkinen, P., Stein, R.F., \& Tuominen, I.\yana{1990}{232}{277}

\bibitem[Brandenburg et al.(2017)]{BPS17}
Brandenburg, A., Petrie, G. J. D., \& Singh, N. K.\yapj{2017}{836}{21}

\bibitem[Caldwell et al.(2017)]{CHK17}
Caldwell, R. R., Hirata, C., \& Kamionkowski, M.\yapj{2017}{839}{91}

\bibitem[Durrer(2008)]{Dur08}
Durrer, R.\ybook{2008}{The Cosmic Microwave Background}
{Cambridge Catalogue, September 2008}

\bibitem[Hughes et al.(2016)]{Hughes}
Hughes, A. L. H., Bertello, L., Marble, A. R., Oien, N. A., Petrie, G., \& Pevtsov, A. A.\arxiv{2016}{1605.03500}

\bibitem[Hurlburt et al.(1984)]{HTM84}
Hurlburt, N. E., Toomre, J., Massaguer, J. M.\yapj{1984}{282}{557}

\bibitem[Hurlburt \& Toomre(1988)]{HT88}
Hurlburt, N. E., \& Toomre, J.\yapj{1988}{327}{920}

\bibitem[Kahniashvili et al.(2014)]{KMLK14}
Kahniashvili, T., Maravin, Y., Lavrelashvili, G., \& Kosowsky, A.\yprd{2014}{90}{083004}

\bibitem[Kamionkowski \& Kosowsky(1999)]{KK99}
Kamionkowski, M., \& Kosowsky, A.\yjour{1999}{Ann. Rev. Nucl. Part. Sci.}{49}{77}

\bibitem[Kamionkowski et al.(1997a)]{Kamion97}
Kamionkowski, M., Kosowsky, A., \& Stebbins, A.\yprl{1997a}{78}{2058}

\bibitem[Kamionkowski et al.(1997b)]{Kamion97b}
Kamionkowski, M., Kosowsky, A., \& Stebbins, A.\yprd{1997b}{55}{7368}

\bibitem[Kamionkowski \& Kovetz(2016)]{KK16}
Kamionkowski, M., \& Kovetz, E. D.\yaraa{2016}{54}{227}

\bibitem[Kandel et al.(2017)]{KLP17}
Kandel, D., Lazarian, A., \& Pogosyan, D.\ymn{2017}{472}{L10}

\bibitem[Kemel et al.(2011)]{KBJ11}
Kemel, K., Brandenburg, A., \& Ji, H.\ypre{2011}{84}{056407}

\bibitem[Krause \& R\"adler(1980)]{KR80}
Krause, F., \& R\"adler, K.-H.\ybook{1980}
{Mean-field Magneto\-hydro\-dy\-na\-mics and Dynamo Theory}{Oxford: Pergamon Press}

\bibitem[Kritsuk et al.(2018)]{Kritsuk18}
Kritsuk, A. G., Flauger, R., \& Ustyugov, S. D.\yprl{2018}{121}{021104}

\bibitem[Moffatt(1978)]{Mof78}
Moffatt, H. K.\ybook{1978}
{Magnetic Field Generation in Electrically Conducting Fluids}
{Cambridge: Cambridge Univ.\ Press}

\bibitem[Park \& Blackman(2012)]{PB12b}
Park, K., \& Blackman, E. G.\ymn{2012}{423}{2120}

\bibitem[Pevtsov et al.(1995)]{PCM95}
Pevtsov, A. A., Canfield, R. C., \& Metcalf, T. R.\yapj{1995}{440}{L109}

\bibitem[Planck Collaboration Int.\ XX(2015)]{PlanckXX}
Planck Collaboration Int.\ XX.\yana{2015}{576}{A105}

\bibitem[Planck Collaboration Int.\ XXX(2016)]{Adam16}
Planck Collaboration Int.\ XXX.\yana{2016}{586}{A133}

\bibitem[Planck Collaboration results XI(2018)]{Akrami18}
Planck Collaboration results XI.\dana{2018}{10.1051/0004-6361/201832618}{1801.04945}

\bibitem[Roberts \& Soward(1975)]{RS75}
Roberts, P. H., \& Soward, A. M.\yan{1975}{296}{49}

\bibitem[Seehafer(1990)]{See90}
Seehafer, N.\ysph{1990}{125}{219}

\bibitem[Seljak \& Zaldarriaga(1997)]{SZ97}
Seljak, U., \& Zaldarriaga, M.\yprl{1997}{78}{2054}

\bibitem[Singh et al.(2018)]{Singh18}
Singh, N. K., K\"apyl\"a, M. J., Brandenburg, A., K\"apyl\"a, P. J., Lagg, A., \& Virtanen, I.\yapj{2018}{863}{182}

\bibitem[Skumanich \& Lites(1987)]{SL87}
Skumanich, A., \& Lites, B. W.\yapj{1987}{322}{473}

\bibitem[Zhang \& Brandenburg(2018)]{ZB18}
Zhang, H., \& Brandenburg, A.\yapjl{2018}{862}{L17}

\bibitem[Zhang et al.(2014)]{ZBS14}
Zhang, H., Brandenburg, A., \& Sokoloff, D. D.\yapjl{2014}{784}{L45}

\bibitem[Zhang et al.(2016)]{ZBS16}
Zhang, H., Brandenburg, A., \& Sokoloff, D. D.\yapj{2016}{819}{146}

\end{thebibliography}

\title{$E$ and $B$ polarizations from inhomogeneous and solar surface turbulence}
\author{Axel Brandenburg$^{1,2,3,4,5}$},
\author{Andrea Bracco$^3$}
\author{Tina Kahniashvili$^{5,6}$}
\author{Sayan Mandal$^{5}$}
\author{Alberto Roper Pol$^{1,7}$}
\author{Gordon J. D. Petrie$^8$}
\author{Nishant K. Singh$^{9}$}

\affil{
$^1$Laboratory for Atmospheric and Space Physics, University of Colorado, Boulder, CO 80303, USA\\
$^2$JILA and Department of Astrophysical and Planetary Sciences, University of Colorado, Boulder, CO 80303, USA\\
$^3$Nordita, KTH Royal Institute of Technology and Stockholm University, Roslagstullsbacken 23, SE-10691 Stockholm, Sweden\\
$^4$Department of Astronomy, AlbaNova University Center, Stockholm University, SE-10691 Stockholm, Sweden\\
$^5$McWilliams Center for Cosmology and Department of Physics,
  Carnegie Mellon University, 5000 Forbes Ave, Pittsburgh, PA 15213, USA\\
$^6$Abastumani Astrophysical Observatory, Ilia State University,
  3-5 Cholokashvili St., 0194 Tbilisi, Georgia\\
$^7$Department of Aerospace Engineering Sciences, University of Colorado,
  Boulder, CO 80303, USA\\
$^8$National Solar Observatory, 3665 Discovery Drive, Boulder, CO 80303, USA\\
$^9$Max-Planck-Institut f\"ur Sonnensystemforschung, Justus-von-Liebig-Weg 3, D-37077 G\"ottingen, Germany
}
\submitted{\today,~ $ $Revision: 1.105 $ $}
%\submitted{Astrophys. J. 836, 21 (2017)}
%\date{Received 2016 October 17; revised 2016 December 26; accepted 2016 December 31; published 2017 February 6}

\begin{abstract}
Gradient- and curl-type or $E$- and $B$-type polarizations have been
routinely analyzed to study the physics contributing to the
cosmic microwave background polarization
and galactic foregrounds.
They characterize the parity-even and parity-odd properties of the
underlying physical mechanisms, such as, for example, hydromagnetic
turbulence in the case of dust polarization.
Here, we study spectral correlation functions characterizing the
parity-even and parity-odd parts of linear polarization for homogeneous
and inhomogeneous turbulence to show that only the inhomogeneous helical
case can give rise to a parity-odd polarization signal.
We also study nonhelical turbulence and suggest that a strong
non-vanishing (here negative) skewness of the $E$ polarization is
responsible for an enhanced ratio of the $EE$ to the $BB$
(quadratic) correlation in both the helical and nonhelical cases.
This could explain the enhanced $EE/BB$ ratio observed recently for
dust polarization.
We close with a preliminary assessment of using the linear polarization of
the Sun to characterize its helical turbulence without being subjected
to the $\pi$ ambiguity that magnetic inversion techniques have to address.
\end{abstract}

\keywords{
Sun: magnetic fields --- dynamo --- magnetohydrodynamics --- turbulence}
\email{brandenb@nordita.org}

\section{Introduction}

Helicity characterizes the swirl of a flow or a magnetic field.
Examples include the cyclonic and anticyclonic flows on a weather map,
which are systematically different in the northern and southern hemispheres.
Similar differences are also seen on the solar surface, where both
flow and magnetic field vectors show swirl.
Both fields play important roles in the solar dynamo, which is believed to be
responsible for the generation of the Sun's large-scale magnetic field
\citep{Mof78,KR80,BS05}.

To determine the solar magnetic helicity, one first needs to determine
the magnetic field $\bb$.
This is done by measuring all four Stokes parameters to compute $\bb$
at the visible surface.
Historically, the first evidence for a helical magnetic field came from
estimates of the mean current helicity density, $\bra{\jj\cdot\bb}$,
where $\jj=\nab\times\bb/\mu_0$ is the current density and $\mu_0$
is the vacuum permeability.
Under the assumption of isotropy, using local Cartesian coordinates,
we have $\bra{\jj\cdot\bb}\approx\bra{j_z b_z}/3$, where
$j_z=(\partial_x B_y-\partial_y B_x)/\mu_0$ is the vertical component of
the current density, which involves only horizontal derivatives.
Another approach is to assume that the magnetic field above the solar
surface is nearly force-free.
A vanishing Lorentz force ($\jj\times\bb=\bm{0}$) implies that $\jj$ is
parallel to $\bb$, so $j_z=\alpha b_z/\mu_0$ with some coefficient $\alpha$.
The sign of $\alpha$ is directly related to the sign of the local current
helicity density.
\cite{See90} and \cite{PCM95} computed $\alpha$ and found it to be
negative in the northern hemisphere and positive in the southern.
This is a statistical result that has been confirmed many times
since then; see, e.g., \cite{Singh18}.

A general difficulty in determining magnetic helicity lies in the fact
that the horizontal magnetic field components are only determined up to
the $\pi$ ambiguity.
In other words, one only measures horizontal magnetic field vectors
without arrow heads.
The actual horizontal field direction is usually ``reconstructed''
by comparing with that expected from a potential magnetic field
extrapolation, which only depends on the vertical magnetic field.
It is unclear to what extent this assumption introduces errors and how
those affect, for example, the scale dependence of the magnetic helicity
that was determined in some of the aforementioned approaches \citep[see,
e.g.,][]{ZBS14,ZBS16,BPS17,Singh18}.

To address the question about the limitations resulting from the $\pi$
ambiguity, we study here another potential proxy of magnetic helicity
that is independent of the $\pi$ ambiguity.
To this end, we decompose Stokes $Q$ and $U$, which characterize linear
polarization, into the $E$ and $B$ polarizations that are routinely
used in the cosmological context \citep{Kamion97,SZ97} as polarized
emission from the cosmic microwave background (CMB).
The $E$ and $B$ polarizations characterize parity-even and parity-odd
contributions; see \cite{KK99} for a review.
In the CMB, correlations of $B$ polarization with parity-even quantities
such as intensity, temperature, or the $E$ polarization
are believed to be proxies of the helicity of the underlying magnetic
field \citep{KMLK14}.

Observations of $E$ and $B\,$ polarizations have been obtained at various
frequencies using the {\em Planck} satellite.
Much of the $B$ polarization is now believed to come from the galactic
foregrounds, including gravitational lensing, for example.
While a definitive $EB$ cross-correlation has not yet been detected,
we do know that the $EE$ correlation is about twice as large as the $BB$
correlation in the diffuse emission \citep{Adam16}.
This was unexpected at the time \citep{CHK17} and will also be addressed in
this paper.

\cite{Kritsuk18} have shown that supersonic hydromagnetic turbulent
star formation simulations are able to reproduce the observed $EE/BB$ ratio
of about two, but the physical reason for this was not identified.
\cite{KLP17} discussed the dominance of magnetic over kinetic energy
density as an important contributor to the enhanced ratio.
Here, instead, we identify a strong skewness of the intrinsic $E$
polarization at all depths along the line of sight as an important
factor.

Our main objective is to study the connection between $EB$ cross-correlation
and magnetic or kinetic helicities in various turbulent flows, which
can be related to what happens at the surface of the Sun.
As we will show in this paper, such a connection exists only
under certain inhomogeneous conditions, such as, for example,
in the proximity of a surface above the solar convection zone.
This automatically gives preference to one over the other viewing
direction of the plane perpendicular to the line of sight.
In homogeneous turbulence, by contrast, there is no way of differentiating
one viewing direction from the other.

\begin{figure}[t!]\begin{center}
\includegraphics[width=\columnwidth]{sketch}
\end{center}\caption[]{
Sketch illustrating the same shape of streamlines
for updrafts and downdrafts for convection in the northern
hemisphere ($\grav\cdot\OO<0$), as viewed from the top down.
}\label{sketch}\end{figure}

Certain physical circumstances may well give preference to one over the
other side of a plane.
The CMB may indeed be one such example.
Rotating stratified convection is another rather intuitive example, as
illustrated in \Fig{sketch}.
In this sketch one sees the flow lines of cyclonic convection around
downflows and upflows in the northern hemisphere as viewed from the top, so
all converging inflows attain a counterclockwise swirl, and all
diverging outflows attain a clockwise swirl.
As seen from the sketch, however, the orientations of both flow patterns
consisting of unsigned flow lines is the same.
This curl-type pattern gives rise to $B$-type
polarization of positive sign \citep[e.g.][]{KMLK14},
as will be verified in the next section.
It is only if we were to flip this plane or
when inspecting this pattern from beneath that we will
see a mirror image of the original pattern and therefore the opposite
sign of the $B$ polarization; see \cite{Dur08}.
Here, the $E$ polarization is the same when viewed from beneath (or in
a mirror).
This gives rise to a systematic $EB$ correlation.
The $E$ polarization will also be positive in this case if a ring-like
pattern dominates over a star-like pattern.
This results in a positive $EB$ correlation
in the north and a negative one in the south.

The purpose of this work is to use various numerical simulations
to determine the relation between magnetic helicity and the $EB$
correlation that is derived from just the horizontal field vectors without
knowledge of which of the two horizontal directions the vector points into
(i.e., the $\pi$ ambiguity).
The simulations include decaying homogeneous helical and nonhelical
turbulence and rotating stratified convection, which can serve as a
prototype for convective turbulence at the solar surface.

\section{$E$ and $B$ polarization}

\subsection{Formalism}

We consider magnetic field vectors, $\bb=(b_x, b_y, b_z)$, in one or
several two-dimensional $xy$ cross sections in a three-dimensional volume.
We assume that this magnetic field affects the polarization of the
electromagnetic radiation whose electric field vectors in the
$(x,y)$ plane are $\ee=(e_x,e_y)$.
It is convenient to use complex notation and write this vector as
$e_x+\ii e_y\equiv|e|\,\exp(\ii\psi_e)$, where $\psi_e$ is the angle
of the electric field with the $x$ axis.
Likewise, we write the magnetic field in the plane, $(b_x,b_y)$,
as $b_x+\ii b_y\equiv|b|\,\exp(\ii\psi_b)$.
For electromagnetic radiation, electric and magnetic field vectors are
at right angles to each other, so $\psi_e=\psi_b+\pi/2$.
In complex form, the intrinsic (or local) polarization $p(x,y,z)$ is
proportional to the square of the complex electric field,
i.e., $p\propto(e_x+\ii e_y)^2\propto-\epsilon\,(b_x+\ii b_y)^2$,
where $\epsilon(x,y,z)$ is the local emissivity.
The magnetic field of the electromagnetic radiation aligns with the
ambient magnetic field, so that
\EQ
p=-\epsilon\,(b_x+\ii b_y)^2/\bb^2.
\EN
In most of the cases, we assume $\epsilon\propto\bb^2$,
which would be appropriate for the Sun \citep{SL87,BDTSMW14}.
For dust polarization, on the other hand, we assume
$\epsilon$ to be independent of $|\bb|$; see
\cite{PlanckXX} and \cite{Bracco18} for details.
The observable complex polarization, $P=Q+\ii U$,
is the line-of-sight integral
\EQ
P(x,y)=\int p(x,y,z)\,e^{-\tau(x,y,z)}\,\dd z,
\label{LOSintegr}
\EN
with $\tau(x,y,z)$ being the optical depth with respect to the observer.
If the medium can be considered optically thin, as for diffuse dust emission
in the interstellar medium, we can set $\tau=0$.
This will also be done in the present work.
In addition, we study the polarization from individual slices,
which corresponds to an optically thick case for that slice.

Next, we define $R=E+\ii B$, where $E$ and $B$ are the parity-even and
parity-odd contributions to the complex polarization, respectively.
They are related to each other in Fourier space via
\citep{Kamion97,Kamion97b,SZ97,KK16}
\begin{equation}
\tilde{R}(k_x,k_y)=(\hat{k}_x-\ii \hat{k}_y)^2\tilde{P}(k_x,k_y),
\end{equation}
where $\hat{k}_x$ and $\hat{k}_y$ are the $x$ and $y$ components
of the planar unit vector $\hatkk=\kk/k$, with $\kk=(k_x,k_y)$ and
$k=(k_x^2+k_y^2)^{1/2}$ being the length of $\kk$.
Tildes indicate Fourier transformation over $x$ and $y$.
We transform $\tilde{R}$ back into real space to obtain $E(x,y)$ and $B(x,y)$
at a given position $z$.
We plot contours of $E$ and $B$ and overplot polarization vectors with angles
\EQ
\chi_E=\half\arg P_E \quad\mbox{and}\quad
\chi_B=\half\arg P_B,
\EN
where $P_E$ and $P_B$ are computed in Fourier space as
$\hat{P}_E=(\hat{k}_x+\ii \hat{k}_y)^2\tilde{E}$ and
$\hat{P}_B=(\hat{k}_x+\ii \hat{k}_y)^2(\ii\tilde{B})$.

We consider shell-integrated spectra along a ring of radius $k=|\kk|$
in wavenumber space of the form
\begin{equation}
C_{XY}(k)=\int_0^{2\pi}\tilde{X}(\kk)\,\tilde{Y}^*(\kk)\,k\,\dd\phi_k,
\label{KKnull}
\end{equation}
where $\phi_k$ is the azimuthal angle in Fourier space, and
$\tilde{X}$ and $\tilde{Y}$ are Fourier transformed fields
that each (or both) stand for $\tilde{E}$ or $\tilde{B}$.
Thus, we consider the spectra $C_{EE}(k)$, $C_{EB}(k)$, and $C_{BB}(k)$.
In some cases, we consider one-point correlators, which are equal to
the integrated spectrum, i.e., $\bra{XY}_{xy}=\int C_{XY}(k)\,\dd k$.
In the following, when we sometimes talk about $EE$ or $BB$ correlations,
we always mean the spectral correlation functions $C_{EE}(k)$ and
$C_{BB}(k)$.

\subsection{Two-scale analysis}
\label{PolarizationAnalysis}

In the Sun, we expect opposite signs of the $EB$ correlation in the
northern and southern hemispheres.
An analogous situation has been encountered previously in connection
with magnetic helicity measurements.
To prevent cancellation from contributions of opposite signs coming
systematically from the two hemispheres, one has to allow for a
corresponding modulation of the sign between the hemispheres.
We refer here to the work of \cite{RS75} for the general formalism in
the context of dynamo theory and to \cite{BPS17} for the application to
observational data similar to those discussed here.
The two-scale formalism has so far only been developed for Cartesian
geometry, but it is conceivable that it can also be extended to spherical
harmonics.

Here, we assume that the $x$ direction points in longitude and the
$y$ direction points in latitude.
To account for a sinusoidal modulation in latitude proportional to
$\sin K_y y$, we compute the following generalized spectrum as
\begin{equation}
C_{XY}(\KK,k)=\int_0^{2\pi}\tilde{X}(\kk+\half\KK)\,\tilde{Y}^*(\kk-\half\KK)\,k\,\dd\phi_k,
\label{KKnull}
\end{equation}
and plot $-\Imag \, C_{XY}(\KK,k)$ versus $k$ for $\KK=(0,K_y)$; see
also \cite{Singh18} and \cite{ZB18} for recent applications.

\subsection{A simple example}

We consider gradient- and curl-type vector fields \citep{Dur08}
\begin{equation}
F_i(x,y)=\partial_i f,\quad
G_i(x,y)=\epsilon_{ij}\partial_j g,
\end{equation}
using
\begin{equation}
f=f_0\cos kx\cos ky,\quad
g=g_0\cos kx\cos ky.
\end{equation}
The two-dimensional projection of an otherwise three-dimensional
magnetic field in this model is given by $\bb(x,y)=\FF+\GG$.
Here, $\epsilon_{ij}$ is the totally antisymmetric tensor in two
dimensions, so $\epsilon_{12}=1$, $\epsilon_{21}=-1$, and zero otherwise.
Assuming $k=1$ in a domain $-\pi<(x,y)<\pi$, we have
\begin{equation}
\FF=-f_0\pmatrix{ \sin x\cos y\cr \cos x\sin y},\quad
\GG= g_0\pmatrix{-\cos x\sin y\cr+\sin x\cos y}.
\end{equation}
In \Fig{p} we show examples of polarization maps for {\em different
combinations} of the coefficients $(f_0,g_0)$.
The polarization vectors correspond to $\bb$ vectors without an arrow.
Pure $E$ polarization occurs whenever either $f_0$ or $g_0$ vanish, whereas
pure $B$ polarization occurs whenever $|f_0|=|g_0|$.
Thus, there is no direct correspondence between gradient- and curl-type
vector fields and gradient- and curl-type polarization.
Thus, Equation~(5.84) of \cite{Dur08} is incorrect
(R.~Durrer 2018, private communication).

\begin{figure}[t!]\begin{center}
\includegraphics[width=\columnwidth]{p}
\end{center}\caption[]{
(a) and (b): pure $E$ polarization for cases
$(f_0,g_0)=(1,0)$ and $(0,1)$, respectively.
(c) and (d): pure B polarization for cases
$(f_0,g_0)=(1,\pm1)$, for upper and lower
signs, respectively.
}\label{p}\end{figure}

\begin{figure}[t!]\begin{center}
\includegraphics[width=\columnwidth]{pscan}
\end{center}\caption[]{
Dependence of real-space correlations
$c_{\rm S}(\phi)$ (top) and $c_{\rm A}(\phi)$ (bottom) on $\phi$.
The inset shows a parametric representation
of $c_{\rm A}(\phi)$ versus $c_{\rm S}(\phi)$.
The open blue and filled red symbols denote the points where
$\bra{E^2}/\bra{B^2}=1$ and $2$, respectively.
The black filled symbols denote the examples of pure $E$ polarizations
in \Figp{p}{a)+(b} and pure $B$ polarizations in \Figp{p}{c)+(d}.
}\label{pscan}\end{figure}

It is convenient to define normalized symmetric and antisymmetric
polarization correlations as
\begin{equation}
c_{\rm S}=\bra{E^2-B^2}\left/\bra{E^2+B^2}\right.
\end{equation}
and
\begin{equation}
c_{\rm A}=2\bra{EB}\left/\bra{E^2+B^2}\right.
\end{equation}
and display them as a function of $\phi$ with
$(f_0,g_0)=(\cos\phi,\sin\phi)$.
Here, angle brackets denote averaging over the $xy$ plane.
The resulting polarization maps are shown in \Fig{pscan}.
In this model, the points in a parametric representation of $c_{\rm A}$
versus $c_{\rm S}$ lie on a closed, nearly circular line; see the inset
of \Fig{pscan}.

Pure $E$ polarization implies $c_{\rm S}=1$, while
pure $B$ polarization implies $c_{\rm S}=-1$.
In both cases, we have $c_{\rm A}=0$.
Furthermore, the case $c_{\rm S}=0$
(which coincides with $c_{\rm A}=\pm1$)
corresponds to $\bra{E^2}/\bra{B^2}=1$.
This is what was theoretically expected in the case of dust polarization
as a probe of ISM turbulence; see \cite{CHK17}.
However, of particular interest is now the case $\bra{E^2}/\bra{B^2}=2$,
which has been detected in foreground polarization with {\em Planck}
\citep{Adam16,Akrami18}.
In our model, this implies $c_{\rm S}=1/3$ with $c_{\rm A}=\pm2\sqrt{2}/3$.

Analogous to the real-space coefficients $c_{\rm S}=0$ and
$c_{\rm A}=0$, we define normalized spectra as
\begin{equation}
c_{\rm S}(k)={C_{EE}(k)-C_{BB}(k)\over C_{EE}(k)+C_{BB}(k)}
\end{equation}
and
\begin{equation}
c_{\rm A}(k)={2C_{EB}(k)\over C_{EE}(k)+C_{BB}(k)}.
\end{equation}
Unless noted otherwise, $E$ and $B$ have been obtained
from simulations through integration along the $z$ direction; see
\Eq{LOSintegr}.

\begin{figure*}[t!]\begin{center}
\includegraphics[width=\textwidth]{peb4}
\end{center}\caption[]{
$E$-mode (left) and $B$-mode (right) polarization for isotropic fully
helical magnetohydrodynamic turbulence using an $xy$ slice of $E$ and $B$
from \cite{BK17} (their Figures~4d--f, for $\Pm=100$).
Dark (light) shades indicate negative (positive) velocity.
In each panel, the insets show an enlarged portion where we also show
the $E$ and $B$ polarization vectors.
}\label{peb4}\end{figure*}

\begin{figure*}[t!]\begin{center}
\includegraphics[width=\textwidth]{pEBmodes_tot}
\end{center}\caption[]{
Spectral correlation functions $c_{\rm S}(k)$ and $c_{\rm A}(k)$ using
line-of-sight integrated polarization (left) and single slice data
(right) computed from decaying isotropic turbulence of \cite{BK17}.
Error margins are indicated in gray.
}\label{pEBmodes_tot}\end{figure*}

\section{Numerical simulations}

\subsection{Isotropic turbulence simulations}
\label{IsotropicTurbulence}

Astrophysical turbulence comes in a multitude of different forms:
it can be helical or nonhelical, it can be magnetically dominated or
subdominant, it can possess cross helicity, with a systematic alignment.
These turbulence simulations provide a means of performing experiments in
a variety of circumstances and environments.
Here, we use three-dimensional simulation data of isotropic MHD turbulence
and consider separately all $xy$ planes.
The simulations describe decaying MHD turbulence with magnetic helicity
in the magnetically dominated case.

In the context of early universe turbulence, we have studied decaying
MHD turbulence, which is magnetically dominated, i.e., the
magnetic energy exceeds the kinetic energy by typically a factor of 10
\citep{BKMRPTV17}.
The turbulence is then mostly driven by the Lorentz force.
The resulting $E$- and $B$-mode polarizations for individual $xy$ slices
are shown in \Fig{peb4} for a particular example.
We avoid using forced turbulence here, because in the helical case the
magnetic field would become bihelical, i.e., it has opposite signs of
magnetic helicity at different wavenumbers \citep{Bra01}.
Instead, we use decaying hydromagnetic turbulence, where, with a helical
initial field, the
sign of magnetic helicity is always the same at all wavenumbers
\citep{KBJ11,PB12b}.
This makes the interpretation of the data more straightforward.
In some cases, we also compare with nonhelical initial fields, but the two
turn out to be rather similar with respect to both the $EE/BB$ correlation ratio
and the $EB$ cross-correlation.

It is interesting to note that, even though the turbulence is nearly fully
helical with a fractional helicity of about 98\%, the $EB$ correlation,
as quantified by $c_{\rm A}(k)$, is actually zero; see \Fig{pEBmodes_tot}.
This was also confirmed for helical magnetic fields threading interstellar
filamentary structures; see the recent work of \cite{Bracco18}.
In hindsight, and as already discussed in the introduction, this is not
surprising because the parity-odd polarization, as measured by the $EB$
cross-correlation characterizes the shape of two-dimensional structures
on a surface---not in a three-dimensional volume.
Thus, it can distinguish between the shapes of the two letters {\sf p}
and {\sf q}, which are mirror images of one another.
In the solar context, one may think of an arrangement of three spots
of different magnetic field strengths on a plane surface.
This arrangement implies a certain sign of magnetic helicity on one
side of the surface, as was recently demonstrated by \cite{BB18}.
In three dimensions, however, we can flip any structure and view it from
the backside, provided both directions are physically equivalent, which
is the case when the system is homogeneous.
The superposition of flipped and unflipped versions results in a vanishing
$c_{\rm A}(k)$.

\begin{figure*}[t!]\begin{center}
\includegraphics[width=\textwidth]{phisto}
\end{center}\caption[]{
Probability density functions of $E$ and $B$ polarization (left) and those of
$\meanE$ and $\meanB$ (right) in semilogarithmic (top) and
linear representations (bottom) for helical turbulence
\citep[the runs shown in Figures~4d--f of][solid lines]{BK17} and
nonhelical turbulence \citep[Run~A of][dashed lines]{BKMRPTV17}.
}\label{phisto}\end{figure*}

\begin{figure*}[t!]\begin{center}
\includegraphics[width=\textwidth]{pEBmodes_tot_dust}
\end{center}\caption[]{
Similar to \Fig{pEBmodes_tot}, but for dust polarization.
}\label{pEBmodes_tot_dust}\end{figure*}

\begin{figure*}[t!]\begin{center}
\includegraphics[width=\textwidth]{phisto_dust}
\end{center}\caption[]{
Similar to \Fig{phisto}, but for dust polarization.
}\label{phisto_dust}\end{figure*}

In \Fig{pEBmodes_tot} we also see that $c_{\rm S}(k)$, based on the
line-of-sight integral in \Eq{LOSintegr}, approaches unity.
In other words, the $EE$ polarization exceeds the $BB$ polarization by
a factor of over a hundred in this case.
This is surprising, because in each of the individual planes, e.g., that
shown in \Fig{peb4}, the $EE$ correlation exceeds the $BB$ correlation
only by a factor of about 2 at $k/k_1\approx30$; see the second panel of
\Fig{pEBmodes_tot}, which was also what was found by \cite{Kritsuk18}
using realistic simulations of supersonic turbulence.
Here and elsewhere, error margins have been computed by using any one-third
of the original data and estimating the error as the largest departure from
the full average.

To understand the reason for this, we must look for the possibility of
excessive and preferential cancellation in $B(x,y)$ compared to $E(x,y)$.
In this connection, we recall that, because the transformation from
$(Q,U)$ to $(E,B)$ is a linear one, the line-of-sight integral in
\Eq{LOSintegr} can also be carried out over $E+\ii B$, which is what
we do when we talk about preferential cancellation in $B$ compared to $E$.

\begin{figure}[t!]\begin{center}
\includegraphics[width=\columnwidth]{phisto_bb}
\end{center}\caption[]{
Histogram of the three components of the magnetic field for the
helical turbulence run.
}\label{phisto_bb}\end{figure}

\begin{figure*}[t!]\begin{center}
\includegraphics[width=.98\textwidth]{pslice}
\end{center}\caption[]{
Magnetic field vectors and line-of-sight component (color-coded; left)
as well as $E$-mode (middle) and $B$-mode polarization (right)
in rotating convection viewed from the top onto the convecting layer.
}\label{pslice}\end{figure*}

\begin{table}[b!]\caption{
Variance, skewness, and kurtosis for the distributions shown
in \Fig{phisto}.
}\vspace{12pt}\centerline{\begin{tabular}{cc|rrrr}
quantity & helical? & $E\;$ & $B\;$ & $\meanE\;$ & $\meanB\;$ \\
\hline
$\sigma$ & yes & $ 0.77$ & $0.63$ & $ 1.51$ & $ 0.25$ \\
         &  no & $ 0.77$ & $0.64$ & $ 1.65$ & $ 0.58$ \\
$\SKEW$  & yes & $-0.55$ & $0.00$ & $-0.45$ & $-0.11$ \\
         &  no & $-0.61$ & $0.00$ & $-0.53$ & $-0.05$ \\
$\KURT$  & yes &   3.09  &  1.54  &   0.56  &   0.58  \\
         &  no &   3.87  &  1.69  &   0.84  &   0.08  \\
\label{Tskew}\end{tabular}}\end{table}

\begin{table}[b!]\caption{
Similar to \Tab{Tskew}, but for the case of dust emission
shown in \Fig{phisto_dust}.
}\vspace{12pt}\centerline{\begin{tabular}{cc|rrrr}
quantity & helical? & $E\;$ & $B\;$ & $\meanE\;$ & $\meanB\;$ \\
\hline
$\sigma$ & yes & $ 0.76$ & $0.65$ & $ 1.21$ & $ 0.56$ \\
         &  no & $ 0.76$ & $0.65$ & $ 1.45$ & $ 0.85$ \\
$\SKEW$  & yes & $-0.19$ & $0.00$ & $-0.19$ & $ 0.01$ \\
         &  no & $-0.20$ & $0.00$ & $-0.24$ & $-0.04$ \\
$\KURT$  & yes & $-0.02$ & $0.26$ & $ 0.07$ & $-0.04$ \\
         &  no & $+0.03$ & $0.24$ & $ 0.16$ & $ 0.00$ \\
\label{Tskew_dust}\end{tabular}}\end{table}

In \Fig{phisto}, we show the probability density functions of $E(x,y,z)$
and $B(x,y,z)$ and compare them with those of the line-of-sight or $z$-integrated
values that we denote here by $\meanE(x,y)$ and $\meanB(x,y)$.
Their variances are $\sigma_E^2=\bra{E^2}-\bra{E}^2$ and
$\sigma_B^2=\bra{B^2}-\bra{B}^2$.
In all cases, the averages are negligible, i.e.,
$\bra{E}^2\approx0$ and $\bra{B}^2\approx0$.
It turns out that, while $B(x,y,z)$ and $\meanB(x,y)$ are symmetric
about zero, $E(x,y,z)$ and $\meanE(x,y)$ are not.
This is quantified by the skewness,
\EQ
\SKEW(E)=\bra{E^3}/\sigma_E^3,\quad
\SKEW(B)=\bra{B^3}/\sigma_B^3.
\EN
These values are listed in \Tab{Tskew} both for helical and
nonhelical turbulence.
These simulations correspond to the runs shown in Figures~4d--f
of \cite{BK17} for the helical case and Run~A of \cite{BKMRPTV17}
for the nonhelical case.
For completeness, we also list there the kurtoses of those fields,
which are defined as
\EQ
\KURT(E)=\bra{E^4}/\sigma_E^4-3,\quad
\KURT(B)=\bra{B^4}/\sigma_B^4-3.
\EN
The consequences of a non-vanishing skewness of $E$ become clear when
looking at the probability density functions of $\meanE(x,y)$ and
$\meanB(x,y)$ in \Fig{phisto}, which show a dramatic difference for
large values where $|E|>\sigma_E$, because now positive and negative
pairs of equal strengths have different abundance or probability and do
not cancel.
The reason for this asymmetry lies in the nature of turbulence, which
has a preference of producing large tails of negative $E$ polarization,
which corresponds to a preference of radial over circular patterns.

In the results presented above, we have assumed that the local emissivity
$\epsilon$ is proportional to $\bb^2$, but this is not realistic in all
astrophysical contexts, such as, for instance, in the case of dust polarization,
which is the case for which an enhanced $EE/BB$ correlation
ratio has been found.
In \Fig{pEBmodes_tot_dust}, we show that for constant $\epsilon$, i.e.,
independent of $|\bb|$, we still find $c_{\rm S}>0$, but it is now no
longer so close to unity as in the case when $\epsilon\propto\bb^2$.
Instead, we have $c_{\rm S}\approx0.6$ for intermediate values of $k$,
which corresponds to $C_{EE}/C_{BB}\approx7$.
The result for individual slices is, however, less strongly affected by
the choice of $\epsilon$.

The corresponding probability density functions are shown in
\Fig{phisto_dust}.
We see that the basic asymmetry of the probability density function of $E$
still persists both for individual slices and for the integrated maps, but
the tails of the distribution are now less extended; see \Tab{Tskew_dust}
for the corresponding values of skewness and kurtosis.
As already explained above, the asymmetry in $E$ results here from a
dominance of circular patterns.
However, even a preference of radial patterns would cause asymmetry, albeit
with the other sign.
Any such asymmetry would always lead to an excess of $EE$ correlations
over $BB$ correlations and hence an enhanced $EE/BB$ ratio.

The relative importance of radial patterns over circular ones
is a qualitatively new property of turbulent motions that needs to be
studied further.
It does not imply any asymmetry in the individual components of the magnetic
field, as shown in \Fig{phisto_bb}.

\subsection{Convection}

Next, we perform hydrodynamic simulations with gravity $\grav=(0,0,-g)$
and angular velocity $\OO=(0,0,\Omega)$ in a layer
$z_{\rm bot}\leq z\leq z_{\rm top}$, heated from below.
Here, $z_{\rm top}-z_{\rm bot}\equiv d$ is the thickness of the layer.
The governing equations for density $\rho$, velocity $\uu$, the specific
entropy $S$, and the magnetic vector potential $\aaaa$ are given by
\begin{equation}
{\DD\ln\rho\over\DD t}=-\nab\cdot\uu,
\end{equation}
\begin{equation}
\rho{\DD\uu\over\DD t}=-\nab P+\rho\grav-2\OO\times\rho\uu
+\jj\times\bb+\nab\cdot(2\nu\rho\SSSS),
\end{equation}
\begin{equation}
\rho T{\DD S\over\DD t}=K\nabla^2 T+\eta\mu_0\jj^2+2\nu\rho\SSSS^2,
\end{equation}
\begin{equation}
{\partial\aaaa\over\partial t}=\uu\times\bb+\eta\nabla^2\aaaa,
\end{equation}
where $P$ is the pressure with $S=c_v\ln P-c_p\ln\rho$, which is defined
up to some additive constant, $c_p$ and $c_v$ are the specific heats at
constant pressure and density, respectively, $T$ is the temperature with
$P/\rho=(c_p-c_v)T$ being the ideal gas equation of state,
$K$ is the thermal diffusivity, $\nu$ is the kinematic viscosity,
$\eta$ is the magnetic diffusivity, $\bb=\bb_0+\nab\times\aaaa$
is the magnetic field with $\bb_0$ being the imposed field, and
$\jj=\nab\times\bb/\mu_0$ is the current density that was already defined
in the introduction.

\begin{table}[b!]\caption{
Parameters for convection simulations.
}\vspace{12pt}\centerline{\begin{tabular}{cccccc}
Run & Ra & $\nu$ & $\lambda$ & $\urms$ & Res. \\
\hline
A &  3600 & 0.01  & 0.35 & 0.050 & $144^2\times48$ \\
B & 14400 & 0.005 & 0.42 & 0.045 & $288^2\times96$ \\
\label{Tconv}\end{tabular}}\end{table}

\begin{figure}[t!]\begin{center}
\includegraphics[width=\columnwidth]{pvarEBmodes_comp}
\end{center}\caption[]{
$c_{\rm A}(k)$ (top) and $c_{\rm S}(k)$ (bottom) for convection
simulations corresponding to the northern (red) and southern (blue)
hemispheres for Runs~A (thin lines) and B (thick lines).
The gray shades indicate error bars obtained from the statistics
over about 50 snapshots covering a time interval of about 1000
time units.
}\label{pvarEBmodes_comp}\end{figure}

We adopt a polytropic stratification with background temperature
$T=-gz/\cp$, so $T=0$ at $z=0$.
We fix $d$ and choose $|z_{\rm top}|/d$ to set the degree of stratification.
The smaller the $|z_{\rm top}|/d$, the stronger is the stratification, i.e.,
the stronger is the temperature contrast.
In the following we choose $|z_{\rm top}|/d=0.1$, so the temperature
changes by a factor of 10; see \cite{HTM84} for a similar setup.
We choose $\grav\cdot\OO$ to be either negative or positive,
corresponding to the northern or southern hemispheres, respectively.
A vertical magnetic field is imposed and tangled by this velocity field.
The simulation setup is similar to that of \cite{HT88}, except that they
did not include rotation, which makes our present simulations therefore
closer to those of \cite{BNPST90}, which did include rotation.

In the following, we denote by $\rho_0$ the density at $z=-d$.
Some of the parameters are listed in \Tab{Tconv}.
The imposed magnetic field points in the $z$ direction
and is given by $B_{0z}=0.02\Beq$, where $\Beq^2=\mu_0\rho_0gd$
is the thermal equipartition field strength.
We use $\Omega=0.2\,(g/d)^{1/2}$ in all cases.
The Rayleigh number is defined as
$(gd^4\rho_{\rm m}/\nu K)\,(\dd s/\dd z)_{\rm m}$, where $\rho_{\rm m}$
and $(\dd s/\dd z)_{\rm m}$ are density and the specific entropy of the
hydrostatic solution in the middle of the domain.

Cross sections of $\bb(x,y)$, $E(x,y)$, and $B(x,y)$ near the surface
are shown in \Fig{pslice} for the results of such a simulation.
One sees cyclonic convection in the northern hemisphere as viewed from
the top, so all converging inflows attain a counterclockwise swirl,
and all diverging outflows have a clockwise swirl.
A similar appearance is also attained by the magnetic field.
If we viewed this pattern from beneath, we
would see as a mirror image of the original pattern and therefore the opposite
sign of the $B$ polarization.
The consequence of this can be seen in \Fig{pvarEBmodes_comp}, where we
plot $c_{\rm S}(k)$ and $c_{\rm A}(k)$ for north (red) and south (blue)
for Runs~A and B, whose parameters are summarized in \Tab{Tconv}.
There is now a systematic $EB$ correlation, so $c_{\rm A}(k)$ is positive
in the north and negative in the south; see \Tab{Tconvhel}.
This is very promising and agrees with our intuition.

\begin{table}[b!]\caption{
Result for convection, as shown in \Fig{pvarEBmodes_comp}.
}\vspace{12pt}\centerline{\begin{tabular}{cc|ccc}
Hemisph.$\!\!$ & $\grav\cdot\OO$ & $c_{\rm A}(k)$ & $\bra{\omega_z u_z}$ & $J_zB_z$ \\
\hline
N & $-$ & $+$ & $-$ & $-$ \\
S & $+$ & $-$ & $+$ & $+$ \\
\label{Tconvhel}\end{tabular}}\end{table}

\begin{figure*}[t!]\begin{center}
\includegraphics[width=\textwidth]{p150416_AR12325}
\end{center}\caption[]{\small{
Solar $E$, $B$, and $EB$ plots in the proximity of AR12325 on 2015--04--16
superimposed on the full-disk image of $E$ polarization.
}}\label{p150416_AR12325}\end{figure*}

In rotating convection in the northern hemisphere, we have $\grav\cdot\OO<0$.
Near the upper surface, a downdraft ($u_z<0$) will suffer a counterclockwise
spin ($\omega_z>0$), so $\omega_z u_z<0$, corresponding to negative
kinetic helicity.
This applies to the sketch shown in \Fig{sketch} (left, for downflows).
Likewise, an updraft ($u_z>0$) will suffer a clockwise
spin ($\omega_z<0$), so again $\omega_z u_z<0$, i.e., the
kinetic helicity is unchanged and its sign equal to that of $\grav\cdot\OO$.
This applies to the sketch shown in \Fig{sketch} (right, for upflows).
Since the polarization vectors have no vector tip, both updrafts and
downdrafts result in the same $E$ and $B$ polarization properties
in each hemisphere.
Therefore $EB$ is positive for $\grav\cdot\OO<0$ (north) and
negative for $\grav\cdot\OO>0$ (south).
In this case, $EB$ does reflect the sign of kinetic helicity,
except that they are opposite to each other.

\section{Prospects of finding solar $EB$ polarization}
\label{SolarData}

We now consider the Stokes $Q$ and $U$ parameters from
the scattering emission on the solar surface.
We ignore Stokes $I$ and $V$ and only look at $Q$ and $U$ at a fixed
wavelength corresponding to the Fe~{\sc I} $630.15\nm$ line
\citep[see][for details on those data]{Hughes}.

\begin{figure*}[t!]\begin{center}
\includegraphics[width=\textwidth]{pspecm_all}
\end{center}\caption[]{\small{
$c_{\rm S}(k)$ and $c_{\rm A}(k)$ for the time average of
all data from 2010 to 2017.
For wavenumbers above $20\Mm^{-1}$, the data have been averaged
over logarithmically spaced bins.
}}\label{pspecm_all}\end{figure*}

An example is shown in \Fig{p150416_AR12325} using data from the
Synoptic Optical Long-term Investigations of the Sun (SOLIS)
instrument of the NSO Integrated Synoptic Program (NISP).
In the following, we analyze the full-disk data such as the one
shown in \Fig{p150416_AR12325}.
In the three insets, we show zoom-ins of $E$, $B$, and the product $EB$
to smaller patch whose location on the solar disk is indicated by a
small square.
In all cases, $E$ and $B$ are computed for the full disk, however.

The resolution of the full-disk data is $2048^2$ pixels, but it turned
out that the spectral power at the highest wavenumbers is rather small.
Therefore, we downsampled the data to a resolution of $512^2$ points
and we verified that no essential information is lost in this process.

To have a chance at finding a definite sign, we separate the
signs in the northern and southern hemispheres by using the two-scale
method discussed in Sect.~\ref{PolarizationAnalysis}.
In \Fig{pspecm_all} we show the result for $c_{\rm S}(k)$ and
$c_{\rm A}(k)$ for the years from 2010 to 2017.
The statistical errors are generally large and there are strong sign
changes for $k<0.03\Mm^{-1}$, suggesting that those values are uncertain.
There is a short range of positive values of $c_{\rm A}(k)$ around
$0.04\Mm^{-1}\la k\la0.1\Mm^{-1}$, but those values are still compatible
with zero within error bars.
This somewhat unexpected result remains subject to further investigations.
As seen from \Tab{Tconvhel}, positive values of $c_{\rm A}(k)$
correspond to negative magnetic helicity, which is expected for
the northern hemisphere and compatible with our two-scale analysis,
where the sign corresponds to that of the northern hemisphere.
The wavenumber interval from $0.04$ to $0.1\Mm^{-1}$ agrees with that
where most of the magnetic power has previously been found from the
SOLIS data; see \cite{Singh18}.
It might therefore be useful to target further work to this wavenumber
range.

We also see that $c_{\rm S}(k)$ is fluctuating around zero.
This shows that the $EE/BB$ correlation ratio is about unity, which
is thus quite different from the {\em Planck} results for dust polarization.
This suggests that the effect of line-of-sight integration discussed in
\Sec{IsotropicTurbulence} is unimportant here and could be a consequence
of the optical thickness being large.

\section{Conclusions}

Our work has identified an important factor governing the enhanced ratio
of $EE$ to $BB$ polarization: a strongly asymmetric $E$ distribution
for helical (nonhelical) turbulence with a skewness of $-0.55$ ($-0.61$)
and $-0.19$ ($-0.20$) for $\epsilon\propto\bb^2$ and $\epsilon=\const$,
respectively, compared with an unskewed $B$ distribution.
This implies that, depending on the extent of the line-of-sight
integration, there will be less cancellation of $E$ compared to $B$,
which explains the enhanced  $EE$ to $BB$ ratio.
This was previously explained in terms of Alfv\'en waves in magnetically
dominated flows \citep{KLP17}.

Under inhomogeneous conditions, the $EB$ cross-correlation is found to
be a meaningful proxy of kinetic and magnetic helicity.
We have shown that such conditions are found in stratified convection
in the presence of rotation.
This became clear from the sketch shown in \Fig{sketch}.
Homogeneous systems, by contrast, are unable to produce any net $EB$
cross-correlation, even if the turbulence is fully helical.
This is because, with respect to a given line of sight,
a helical eddy can face the observer at different viewing angles, where
$B$ can attain positive and negative values, depending on which side of the
plane the observer is facing, while the $E$ polarization can be similar
in both cases, independently of the viewing angle.
For convection, on the other hand, owing to inhomogeneity, it is
impossible to find a local plane whose statistical $EB$ correlations
agree with one that is flipped, so there can be no cancellation.
This was demonstrated by our numerical experiments, which show a dependence
of the $EB$ correlation on the sign of $\grav\cdot\OO$, and thus on the
kinetic and magnetic helicities.

To assess the prospects of determining parity-odd polarization from
solar scattering emission, we have employed the two-scale analysis
to the oppositely helical contributions from north and south.
Unfortunately, a clear antisymmetric spectral correlation could
not be determined as yet.
Even in the $k$ range between $0.04$ and $0.1\Mm^{-1}$,
where most of the magnetic energy is known to reside in the
SOLIS measurements \citep{Singh18}, the positive values obtained
for $c_{\rm A}(k)$ are compatible with zero.
One reason for this poor hemispheric distinction could be that not all
corrections applied to the final vector spectromagnetograph magnetic
field data are included in the spectral data cubes for Stokes $I$, $Q$,
$U$, and $V$ available from the SOLIS website.
This issue needs to be investigated in future work.

\acknowledgements
We thank the referee for a careful assessment of the paper
and for useful comments.
This work has utilized SOLIS data obtained by the NSO Integrated Synoptic
Program (NISP), managed by the National Solar Observatory, which is
operated by the Association of Universities for Research in Astronomy
(AURA), Inc.\ under a cooperative agreement with the National Science
Foundation.
{\em SDO} is a mission for NASA's Living With a Star program.
We acknowledge partial support from the University of Colorado through
its support of the George Ellery Hale visiting faculty appointment,
the National Science Foundation Astrophysics and Astronomy Grant Program
grants AST1615940 \& AST1615100, and the Swiss NSF SCOPES grant IZ7370-152581.
We acknowledge the allocation of computing resources
provided by the Swedish National Allocations Committee at the Center for
Parallel Computers at the Royal Institute of Technology in Stockholm.
This work utilized the Janus supercomputer, which is supported by the National
Science Foundation (award No.\ CNS-0821794), the University of Colorado
Boulder, the University of Colorado Denver, and the National Center for
Atmospheric Research. The Janus supercomputer is operated by the University of
Colorado Boulder.

%r e f


\vfill\bigskip\noindent\tiny\begin{verbatim}
$Header: /var/cvs/brandenb/tex/sayan/EBmodes/paper.tex,v 1.105 2018/12/21 19:19:40 brandenb Exp $
\end{verbatim}

\end{document}

